Author’s response to reviews

Title: HP1-gamma expression is elevated in prostate cancer and is superior to Gleason score as a predictor of biochemical recurrence after radical prostatectomy

Authors:

Jon A Slezak (jaslezak@wisc.edu)
Matthew Truong (mtruong@wisc.edu)
Wei Huang (whuang23@wisc.edu)
David F Jarrard (jarrard@urology.wisc.edu)

Version: 3 Date: 10 February 2013

Author’s response to reviews:

We have now modified our ethics committee statement upon your request.